

# Dependent default times with mixed hitting-time models Mikael Escobar-Bach, Salima Helali, Hajer Rejeb

# ▶ To cite this version:

Mikael Escobar-Bach, Salima Helali, Hajer Rejeb. Dependent default times with mixed hitting-time models. 2024. hal-04853879

# HAL Id: hal-04853879 https://hal.science/hal-04853879v1

Preprint submitted on 22 Dec 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# Dependent default times with mixed hitting-time models

Mikael ESCOBAR-BACH<sup>1</sup>, Salima HELALI<sup>2,3†</sup> and Hajer REJEB<sup>3‡</sup>

<sup>1</sup>Univ Angers, LAREMA, F-49000 Angers, France <sup>2</sup>Univ Paris-Est Créteil, LAMA, F-94010 Créteil, France <sup>3</sup>Univ Sfax, LPS, T-3000 Sfax, Tunisia

#### Abstract

Mixed hitting-time models allow the analysis of competing risks through optimal stopping decisions interpreted as crossing times of latent Lévy processes with heterogeneous thresholds. In this paper, we consider a bivariate time model with dependent default, where observation times are subject to censoring and share a common latent process given by a Lévy subordinator. We establish the identifiability of the model and propose different estimators for the marginal distributions and the joint survival distribution. We establish their asymptotic properties and evaluate the finite-sample performance of our results through a simulation study on synthetic data, followed by an application using real data.

Key Words: dependent risk, mixed hitting-time model, Lévy process.

## 1 Introduction

Threshold-crossing models for duration analysis have recently received considerable attention in various fields, including finance, reliability engineering and survival analysis. From a practical perspective, these models are especially useful in economics, with real applications such as patent races [48, 14], technology adoption [47, 28] and smoking cessation among married couples in socioeconomic settings contexts [4]. The fundamental premise of the mixed hitting-time (MHT) models is to examine the time at which a process crosses a predetermined or random threshold, leading to significant events such as defaults, failures or other critical occurrences. This approach provides a flexible description of survival mechanisms [38, 2] in the context of optimal stopping time problems, where typical solutions are usually defined by threshold-crossing rules [31, 33, 3]. To some extent, it also includes mixed proportional hazards models [39, 34], which are a well-known class of competing risk models with proportional hazard rates. In this paper, we consider a bivariate competing risks model based on the threshold-crossing duration framework, where the two durations correspond to the first-passage times of two dependent latent stochastic processes. The durations T and C are defined as the times when latent processes  $(M_{1,t})_{t\geq 0}$  and  $(M_{2,t})_{t\geq 0}$  exceed independent random barriers  $E_1$  and  $E_2$  respectively. This definition is often referred to as the canonical construction of default times [9, 51] and describes the standard intensity-based approach for modeling defaults. Specifically, we consider two proper cumulative hazard functions  $\Lambda_1$  and  $\Lambda_2$ ,

<sup>\*</sup>e-mail adress: mikael.escobar-bach@univ-angers.fr

<sup>&</sup>lt;sup>†</sup>e-mail adress: helali.salima@gmail.com

<sup>&</sup>lt;sup>‡</sup>e-mail adress: hajer.rejeb06@gmail.com

that is, positive increasing functions on  $\mathbb{R}_+$  such that  $\Lambda_1(0) = \Lambda_2(0) = 0$ . The default times are then defined by

$$T = \inf \{t \ge 0; \ M_{1,t} \ge E_1\}, \quad C = \inf \{t \ge 0; \ M_{2,t}(t) \ge E_2\}$$
(1.1)

and

$$M_{1,t} = \Lambda_1(t) + Y_t, \quad M_{2,t} = \Lambda_2(t) + Y_t \tag{1.2}$$

where  $(Y_t)_{t>0}$  is a right-continuous and increasing Lévy process, also referred to as a subordinator. The intuition behind the choice in (1.2) is that the term structure of the survival probability is modeled through an individual deterministic component, represented by the cumulative hazard functions, and a stochastic component that affects all elements simultaneously. By incorporating a mutual latent stochastic process, this model captures the possibility of both durations ending at the same time, as the two dependent processes can simultaneously jump over their respective thresholds. This property is not common to all competing risk models, but it is also shared with the bivariate distributions of the Marshall-Olkin type. Under generalized Marshall-Olkin (GMO) models, the simultaneous termination of both durations is driven by a common independent shock, as assumed in [46, 43, 27]. GMO models typically assume shocks with immediate effects; however, practical experience shows that such scenarios do not always occur, as shocks may have delayed effects. For example, during a financial crisis, weaker institutions may be affected first, while stronger institutions experience the effects later [see, e.g. 13]. Specific instances of these models originate from a particular dynamic default model proposed for valuing certain financial products [59]. In MHT models, the stochastic processes represent the accumulation of implicit shocks which occur only up to a certain threshold. This means that the observable results of these shocks, particularly when a failure occurs, are not immediate. Rather, they depend on the progression of  $(Y_t)_{t\geq 0}$ , which allows for a more nuanced view of risk dynamics.

In a dependent and bivariate competing risks framework, individuals experience two types of durations, but only the first to occur is observed. In survival analysis, this issue is referred to as dependent censoring and occurs when only a lower bound of the event of interest is returned. In such analyses, either the time to event T or the time to censoring C is observed for each individual in the sample. For instance, participants in clinical trials often withdraw prematurely if they find the drug to be ineffective or experience adverse effects [50]. In credit risk analysis, censoring occurs when a specific loan is being repaid at the moment of data collection [44]. Ignoring this dependence can lead to biased survival estimations, which explains why most existing statistical survival studies assume the independence of censoring. However, this assumption is challenging to verify in practice as the available data typically provide only marginal distributions. Consequently, determining whether survival is truly independent of censoring remains a challenging task, and non-parametric identifiability of the joint model is impossible without prior knowledge of the dependence structure [56]. In this context, adjusting for dependent censoring may require practitioners to have knowledge of the true dependence structure between T and C [see, e.g. 61, 10, 15, 18, 26]. For most approaches, the dependence structure is characterized by copula functions, which capture dependencies between random variables independently of their marginal distributions. However, this requirement presents significant challenges in practical applications, as specifying the correct copula is often difficult. In addition, misspecified copulas may exacerbate model bias, leading to incorrect inferences and misleading results. In survival analysis, several recent advances have addressed the limitation of assuming a known copula by exploring parametric modeling approaches in [17] and semi-parametric models in [22]. Identification results for competing risks models have also been studied with several contributions under different sets of restrictions [see, e.g. 30, 1, 41, 42, 44].

This study focuses on the analysis of dependent risk models based on the canonical construction of default models (1.1). It first addresses the issue of non-parametric identification of the dependence structure as examined in [32]. The identifiability and the estimation of the survival distribution and dependence structure of the proposed model are indeed challenging problems. Studies on structural threshold-crossing models driven by latent stochastic processes are relatively limited in the literature. In [49], the authors examined a competing risk model driven by a latent compound Poisson process, but no formal identification results exist for such a complex structural model. Without censoring, the study in [3] derived non-parametric identification results for spectral negative Lévy processes crossing a heterogeneous threshold. In [44, 27], the non-parametric identifiability of the models is addressed, although these methods require the information on whether  $T \leq C$  and  $T \geq C$ for each observation. In this regard, we demonstrate that our model is identified under a standard censoring model. The main advantage of our approach, compared to existing identifiability results, is that it avoids functional form restrictions on the marginal distributions, maintaining their non/semi-parametric nature, while the structure of dependence does not need to be known or assumed. We show that it is possible to apply commonly used statistical methods from survival analysis to estimate the marginal distributions of T and C as well as the joint survival distribution of (T, C). The remainder of the paper is organized as follows. Section 2 presents the model and the notation. In Section 3, we introduce the main characteristics of the survival copula and its asymptotic properties. Section 4 contains the identification result. A non/semi-parametric estimation of the survival distributions, the joint survival distribution, the Kendall's tau and their asymptotic properties are established in Section 5. Section 6 presents simulation results that confirm the theoretical findings, first through simulations (Subsection 6.1) and then through the use of a real data set (Subsection 6.2). Some relevant concluding remarks and potential perspectives are presented in Section 7, while Section 8 includes the proofs of the theoretical results.

## 2 Model and notations

A Lévy process is a stochastic process with stationary and independent increments that starts at zero and has paths that are continuous from the right. Due to these properties, each such process  $(Y_t)_{t\geq 0}$  is characterized by its marginal Lévy exponent  $\kappa$ , which is related to its Laplace-Stieltjes transformation  $\psi_{Y_t}$  by

$$\psi_{Y_t}(x) = e^{-t\kappa(x)}, \qquad t \ge 0.$$

The Lévy exponent  $\kappa$  fully characterizes the process distribution and has the key properties of being non-decreasing, and for all  $x \geq 0$ , satisfying  $\kappa(x) \leq x\kappa(1)$ . In this work, we consider that the lifetimes of interest defined by the random times T and C are affected by the defaults of a Lévy process as defined in (1.1) where  $E_1$  and  $E_2$  are independent unit exponential random variables. In the sequel, we will also use the following equivalent formulation

$$T = \inf\{t \ge 0; e^{-M_{1,t}} \le U_1\}$$
 and  $C = \inf\{t \ge 0; e^{-M_{2,t}} \le U_2\},\$ 

where  $U_1 = \exp(-E_1)$  and  $U_2 = \exp(-E_2)$ . Note that  $U_1$  and  $U_2$  are uniformly distributed. Based on a right-censoring approach, only the smallest default time is observed. Formally, the outcomes are limited to the couple  $(Z, \delta)$ , defined by

$$Z = \min(T, C) \quad \text{and} \quad \delta = \mathbb{1}_{\{T \le C\}}.$$

The survival functions of the random time Z, T and C are denoted by  $\overline{H}$ ,  $\overline{F}_T$  and  $\overline{F}_C$  respectively. According to Theorem 2.2 in [32],  $\overline{F}_T$  and  $\overline{F}_C$  are given, for any  $t \ge 0$ , by

$$\overline{F}_T(t) = e^{-\Lambda_1(t) - t\kappa(1)}$$
 and  $\overline{F}_C(t) = e^{-\Lambda_2(t) - t\kappa(1)}$ 

To fully characterize the joint distribution of T and C, we apply Sklar's theorem [54], which asserts that for any joint distribution function, there exists a unique copula C, defined on  $F_T(\mathbb{R}_+) \times F_C(\mathbb{R}_+)$ , that captures the dependence between T and C. Specifically, for any  $t, s \ge 0$ , the joint survival probability P is given by

$$P(t,s) = \mathbb{P}(T \le t, C \le s) = \mathcal{C}(F_T(t), F_C(s)).$$

In reliability studies, although this does not change the analysis, it is often more convenient to express a joint survival function in terms of its marginal survival functions, such that

$$\widetilde{P}(t,s) = \mathbb{P}(T > t, C > s) = \widetilde{C}\left(\overline{F}_T(t), \overline{F}_C(s)\right), \qquad (2.1)$$

where the function  $\tilde{C}$  is called the *survival copula*. In the sequel, we show that the identifiability and estimation of the distribution of T and the joint distribution P are feasible for this model. Building on this, we introduce the generalized inverses of the cumulative distribution functions  $\overline{F}_T$  and  $\overline{F}_C$ , denoted by  $\overline{F}_T$  and  $\overline{F}_C$ , respectively, which are defined as

$$\overline{F}_T^-(t) = \inf \left\{ x \ge 0, \ \overline{F}_T(x) \le t \right\} \quad \text{and} \quad \overline{F}_C^-(t) = \inf \left\{ x \ge 0, \ \overline{F}_C(x) \le t \right\}.$$

Recall that for any general distribution function F, the corresponding cumulative hazard function is defined as

$$\Lambda(t) = \int_0^t \frac{F(du)}{\overline{F}(u^-)}, \quad t \ge 0,$$

where  $\overline{F} := 1 - F$  and  $\overline{F}(u^{-}) = \lim_{s \downarrow u} \overline{F}(s)$ . Conversely, the relationship between a distribution function F and its cumulative hazard function  $\Lambda$  can be expressed as

$$\overline{F}(t) =: \exp\left(-\widetilde{\Lambda}(t)\right) = \exp\left(-\Lambda^{c}(t) + \sum_{s \leq t} \log(1 - \Delta\Lambda(s))\right),$$

where  $\Lambda^c$  and  $\Delta\Lambda$  respectively denote the continuous and discontinuous parts of  $\Lambda$  [see p.898 in 52]. Thus, the model can be represented using the survival functions

$$\overline{F}_T(t) =: \exp(-\widetilde{\Lambda}_T(t)) \quad \text{and} \quad \overline{F}_C(t) =: \exp(-\widetilde{\Lambda}_C(t)),$$

where  $\Lambda_T$  and  $\Lambda_C$  define the cumulative hazard functions of T and C, respectively.

# 3 Copulas and dependence properties

In this section, we present several fundamental properties of the proposed model, with a particular focus on the copula function  $\tilde{C}$  and its relationship with the GMO copula function. In [32], a general form for the survival GMO copula is derived and used to define a large class of copulas. In line with this work, we derive the joint survival function and the survival copula associated with the proposed model in the following proposition.

**Proposition 1.** The joint survival function  $\widetilde{P}$  of (T, C) defined in (1.1) is given, for any  $t, s \ge 0$ , by

$$\widetilde{P}(t,s) = \overline{F}_C(s)\overline{F}_T(t)\min\left(\overline{F}_T(t)^{-\alpha(t)}, \overline{F}_C(s)^{-\beta(s)}\right),$$

where

$$\alpha(t) = \frac{t\xi}{\Lambda_1(t) + t\kappa(1)}, \qquad \beta(s) = \frac{\xi s}{\Lambda_2(s) + s\kappa(1)},$$

and  $\xi = 2\kappa(1) - \kappa(2)$ . In particular, the survival copula  $\widetilde{\mathcal{C}}$  is given by

$$\widetilde{\mathcal{C}}(u,v) = uv \min\left(u^{-\alpha\left(\overline{F}_{T}(u)\right)}, v^{-\beta\left(\overline{F}_{C}(v)\right)}\right), \quad (u,v) \in \overline{F}_{T}(\mathbb{R}_{+}) \times \overline{F}_{C}(\mathbb{R}_{+}).$$
(3.1)

Note that the minimum term in (3.1) represents the deviation of the dependence structure from the independent scenario, which occurs when  $\xi = 0$ , and from the complete dependence case, which occurs when  $\alpha \equiv \beta \equiv 1$ . In a broader context, constant functions  $\alpha$  and  $\beta$  are inherently connected to the MO model. The sub-density functions and the hazard functions are provided in the following proposition.

**Proposition 2.** The survival function  $\overline{H}$  of Z is given by  $\overline{H}(t) = e^{-\Lambda_1(t) - \Lambda_2(t) - t\kappa(2)}$  for  $t \ge 0$ . Moreover, the sub-densities  $f_{Z,\delta=0}$  and  $f_{Z,\delta=1}$  of  $(Z,\delta)$  are expressed by

$$f_{Z,\delta=0}(t) = \overline{H}(t)[\Lambda_2'(t) + \kappa(2) - \kappa(1)] \quad and \quad f_{Z,\delta=1}(t) = \overline{H}(t)\left[\Lambda_1'(t) + \kappa(1)\right].$$
(3.2)

In particular, the hazard functions  $\lambda_T$  and  $\lambda_C$  respectively associated with the random variables T and C are given by  $\lambda_T = f_{Z,\delta=1}/\overline{H}$  and  $\lambda_C = f_{Z,\delta=0}/\overline{H} + \xi$ .

Notably, the hazard function of T maintains the same form as in the case of independent censoring. This property ensures that the Nelson-Aalen estimator remains consistent even under the dependent censoring induced by the model construction. Further details on estimating the survival distribution function are provided in Section 5. To conclude this section, we present the following statement, which establishes the connection between the proposed model and the GMO model.

**Proposition 3.** The model defined by (1.1)-(1.2) is a special case of the GMO model. Specifically, we have

$$(T,C) \stackrel{d}{=} (\min(X_1,X_3),\min(X_2,X_3))$$

where  $\stackrel{d}{=}$  denotes equality in distribution,  $X_1, X_2$  and  $X_3$  are three independent random variables such that  $X_3$  follows an exponential distribution with parameter  $\xi$ . Moreover, the cumulative hazard functions  $\Lambda_{X_1}, \Lambda_{X_2}$  of  $X_1$  and  $X_2$ , respectively, are given by:

$$\begin{cases} \Lambda_{X_1}(t) = \Lambda_1(t) + t\kappa(1) - \xi t \\ \Lambda_{X_2}(t) = \Lambda_2(t) + t\kappa(1) - \xi t. \end{cases}$$

It is evident that the simultaneous failure of both T and C arises from the common failure source  $X_3$ . Although the proposed model shares the same distribution as a specific case of the GMO model, it is important to emphasize that our approach introduces key distinctions in modeling dependent censoring. MHT models offer greater flexibility by capturing more complex, time-varying dependencies through stochastic processes. This enables alternative interpretations of real-world censoring mechanisms, such as recurrent events or latent factors. In contrast, GMO models lack a dynamic temporal component and are typically limited to shock-induced dependencies that occur instantaneously, which makes it challenging for them to effectively represent these time-varying dependencies. Additionally, GMO models are constrained by their rigid assumptions of shared shocks, which limits their ability to reflect nuanced dependency structures. MHT models also offer better adaptability in capturing latent factors and unobserved competing risks. Their flexibility also facilitates model estimation, particularly in settings with latent competing risks, where GMO models may be too rigid or face identifiability challenges. Finally, it is often more practical to summarize the complexity of the model using a universal measure of dependence, such as Kendall's tau coefficient.

**Remark 1.** Based on Proposition 2 in [27], if  $\Lambda_1$  and  $\Lambda_2$  are strictly increasing, the Kendall's tau corresponding to the proposed model is given by

$$\tau = 2\xi \int_0^{+\infty} \overline{H}(u)^2 du.$$

## 4 Identifiability

In this section, we investigate the model's identifiablity. Recall that a statistical model  $\mathcal{P} = (P_{\gamma})_{\gamma \in \Gamma}$ is said to be identifiable if the map  $\gamma \to P_{\gamma}$ , defined on  $\Gamma$ , is injective. In our context, identifiability implies that the vector  $(\Lambda_1, \Lambda_2, \kappa(1), \kappa(2))$  uniquely specifies the density of the observable random vector  $(Z, \delta)$ . That is,

if 
$$f_{Z,\delta=.,\Gamma_1} = f_{Z,\delta=.,\Gamma_2}$$
 then  $\Gamma_1 = \Gamma_2$ ,

where  $\Gamma_i = (\Lambda_1^i, \Lambda_2^i, \kappa(1)^i, \kappa(2)^i)$ , for i = 1, 2. Examining the identifiability with dependent censoring is no simple task, particularly when considering the dependence structure of the survival copula  $\widetilde{\mathcal{C}}$ . This is due to the right censoring mechanism, which prevents the simultaneous observation of the pair (T, C), making it difficult to directly discern the relationship between T and Cfrom the observed data. In the following theorem, we prove the identifiability of the proposed model.

## **Proposition 4.** Assuming that $\Lambda'_1(0) = \Lambda'_2(0) = 0$ , the model defined by (1.1)–(1.2) is identifiable.

Unlike the identifiability results in semi-parametric models discussed in [17, 21, 22] or in the MHT models in [3, 44], our approach does not require complex assumptions about the marginal distributions or the joint model to establish the model identifiability. Furthermore, compared to the model identification in [27], we do not require any additional information beyond the censoring status  $T \leq C$ , which is typically available in standard survival analysis. The only necessary assumption concerns the regularity of the marginal densities  $\Lambda_1$  and  $\Lambda_2$ , which is satisfied by a wide range of cumulative hazard functions. In particular, this preserves the semi-parametric character of the model and does not impose any functional form restrictions on the marginal distributions. Overall, the identifiability of our model suggests that we can determine the relationship between T and C, observing only their minimum through the censored data  $(Z, \delta)$ .

### 5 Estimation

We now consider the estimation of the survival distributions  $\overline{F}_T$  and  $\overline{F}_C$ , as well as the joint survival  $\widetilde{P}$  of the survival time T and the censoring time C. For this, we assume that we have an independent and identically distributed (i.i.d.) sample  $(Z_i, \delta_i)_{1 \le i \le n}$  drawn from the observed model  $(Z, \delta)$ .

## 5.1 Non-parametric estimation of $\overline{F}_T$

According to Equation (3.2) in Proposition 2, it turns out that the Nelson-Aalen estimator  $\Lambda_{T,n}$  of the cumulative hazard function  $\Lambda_T$  remains consistent, unlike most dependent censoring models. The Nelson-Aalen estimator of  $\Lambda_T$  is then defined by

$$\Lambda_{T,n}(t) = \sum_{i:Z_{(i)} \le t} \frac{d_i}{n-i+1} \quad \text{with} \quad d_i = \sum_{j=1}^n \mathbb{1}_{\{Z_j = Z_{(i)}, \delta_j = 1\}} \quad \text{and} \quad t > 0,$$
(5.1)

where  $Z_{(i)}$  is the *i*-th order statistic among the *n* survival time observations. According to Lemma 1 of [11], we also conclude that the Kaplan-Meier (KM) estimator  $\overline{F}_{T,n}$  defined by  $\overline{F}_{T,n} = \exp(-\Lambda_{T,n})$  provides a consistent estimator of the survival function  $\overline{F}_T$ . Note that its asymptotic properties can then be easily derived from established results on the KM estimator under a survival model with independent censoring (see e.g. [6, 45]).

# 5.2 Semi-parametric estimation of $\overline{F}_C$ , $\widetilde{P}$ and $\tau$

Since the survival estimation  $\overline{F}_T$  can be estimated by the KM estimator, we need to approximate the survival function  $\overline{F}_C$  and the parameter  $\xi$  in order to estimate the survival joint distribution. To this end, we consider a semi-parametric approach based on maximum likelihood (ML) with Bernstein polynomial functions. Bernstein polynomials have been used by many authors in smoothing estimators (see e.g. [58], [35] and [40]) using functions of the form

$$s \in [0,1] \to \sum_{k=0}^{m} \phi_k b_{k,m}(s), \quad m \in \mathbb{N}$$

where *m* is the degree,  $b_{k,m}(s) = C_m^k s^k (1-s)^{m-k}$  and  $\overline{\phi}_m = (\phi_0, \dots, \phi_m)^T \in \mathbb{R}^{m+1}$ . In this context, we assume that both random times *T* and *C* have equal supports. For compact supports  $S_{F_C} = S_{F_T} = [a, b]$ , we can transform the data linearly to the interval [0, 1], using

$$\widetilde{Z} = (Z-a)/(b-a) = \min(T-a)/(b-a), (C-a)/(b-a)) = \min(\widetilde{T}, \widetilde{C}).$$

If  $F_{\widetilde{C},m}$  and  $g_{\widetilde{C},m}$  denote estimates of the distribution and the density functions of  $\widetilde{C}$ , respectively, then estimators of the function  $F_C$  and  $f_C$  can be obtained using the following relations:

$$F_{C,m}(t) = F_{\widetilde{C},m}\left(\frac{t-a}{b-a}\right)$$
 and  $f_{C,m}(t) = \frac{1}{b-a}f_{\widetilde{C},m}\left(\frac{t-a}{b-a}\right)$ ,

for  $t \in [a, b]$ . Transformations such as  $\widetilde{Z} = Z/(1+Z)$  and  $\widetilde{Z} = (1/2) + (\pi/2) \tan^{-1}(Z)$  can be used also to handle the cases of random variables with support  $\mathbb{R}_+$  and  $\mathbb{R}$ , respectively, with the proper change of variables to recover  $F_{C,m}$  and  $f_{C,m}$ . In the following, we consider that  $S_{F_C} = S_{F_T} = [0, 1]$ . Based on Proposition 2, the log-likelihood is given by

$$\mathcal{L}(\xi, \overline{F}_C, f_C) = \sum_{i=1}^n \delta_i \log\{f_{Z,\delta=1}(Z_i)\} + (1 - \delta_i) \log\{f_{Z,\delta=0}(Z_i)\}$$
  
= 
$$\sum_{i=1}^n \delta_i \log\{f_Z(Z_i) - f_{Z,\delta=0}(Z_i)\} + (1 - \delta_i) \log\{f_{Z,\delta=0}(Z_i)\},$$

where  $f_{Z,\delta=0}(t) = \overline{F}_T(t)e^{t\xi} \left[ f_C(t) - \xi \overline{F}_C(t) \right]$  and  $f_Z$  is the density of Z. The direct maximization of this likelihood can be challenging, since it depends on an unknown functions  $f_Z$  and  $\overline{F}_T$ . The idea is now to consider an approximation of the log-likelihood  $\mathcal{L}$  through the following steps: (i) we approximate  $f_Z$  using a non-parametric kernel method

$$f_{Z,n}(t) = \frac{1}{nh} \sum_{i=1}^{n} K\left(\frac{t-Z_i}{h_n}\right), \qquad (5.2)$$

where K represents a kernel function and  $h_n > 0$  is a bandwidth parameter.

- (ii) the function  $\overline{F}_T$  is estimated using the KM estimator  $\overline{F}_{T,n}$ .
- (iii) the functions  $F_C$  and  $f_C$  are approximated using the Bernstein polynomials. The estimators for the distribution function and the density function are then given by

$$F_{C,m}(t|\overline{\phi}_m) = \sum_{k=0}^m \phi_k b_{k,m}(t) \text{ and } f_{C,m}(t) = \sum_{k=0}^m \phi_k b'_{k,m}(t),$$
(5.3)

where the coefficients  $\overline{\phi}_m$  are replaced by estimates obtained from the ML approach.

Therefore, we approximate and parameterize the distribution  $F_C$  using a mixture of beta distributions, and we estimate  $\overline{\phi}_m$  as parameters of the log-likelihood function  $\mathcal{L}$ . This approximation is motivated by the uniform convergence of  $F_{C,m}$  as provided in Theorem 2.1 of [8], as well as by, common results on Bernstein polynomial estimation, which ensure that the best convergence rate of  $F_{C,m}(t)$  to  $F_C(t)$  is of order  $m^{-1}$ .

In summary, let  $\mathcal{L}_m(\xi, \overline{\phi}_m)$  denote the approximate log-likelihood obtained by replacing  $\overline{F}_T$ ,  $f_Z$ ,  $\overline{F}_C$  and  $f_C$  with  $\overline{F}_{T,n}$ ,  $f_{Z,n}$ ,  $\overline{F}_{C,m}$  and  $f_{C,m}$  respectively. We consider the parameter vector  $\theta = (\overline{\phi}_m, \xi)$  and define the ML estimator

$$\theta_n = \underset{\theta \in \mathbb{R}^{m+2}}{\operatorname{argmax}} \mathcal{L}_m(\theta).$$

Based on the estimator  $\theta_n = (\overline{\phi}_{n,m}, \xi_n)$ , we obtain the estimators of the survival function  $\overline{F}_C$  and the joint law  $\widetilde{P}$  namely

$$\overline{F}_{C,m,n}(t) = 1 - \sum_{k=0}^{m} \phi_{k,n} b_{k,m}(t),$$
$$\widetilde{P}_{m,n}(t,s) = \overline{F}_{C,m,n}(s) \overline{F}_{T,n}(t) \min\left(\overline{F}_{T,n}(t)^{-\alpha_n(t)}, \overline{F}_{C,m,n}(s)^{-\beta_{m,n}(s)}\right).$$

where

$$\alpha_n(t) = \frac{\xi_n t}{-\ln(\overline{F}_{T,n}(t))} \quad \text{and} \quad \beta_{m,n}(t) = \frac{\xi_n t}{-\ln(\overline{F}_{C,m,n}(t))}$$

Finally, we propose an estimator for the Kendall's tau coefficient based on the above expression. Conveniently, this coefficient depends only on the survival function of the random variable Z and the stochastic process parameter  $\xi$ . Using the ML estimate of the former parameter, we obtain the estimator

$$\tau_n = 2\xi_n \int_0^{+\infty} \overline{H}_n(u)^2 du.$$

#### 5.3 Asymptotic properties

This section presents the consistency and asymptotic normality of the estimators defined in Subsection 5.2, along with the set of assumptions required to prove these results. To establish the asymptotic normality of  $\theta_n$ , we rely on the framework provided by [60] which develops sufficient conditions for the asymptotic normality of the maximum likelihood estimators under misspecified models. To do so, we consider the optimization problem based on the minimization of the Kullback-Leibler information criterion given by  $I(g, h; \theta) = \mathbb{E} (\log [g(Z, \delta; \xi)/h(Z, \delta; \theta)])$  with densities

$$g(t,b;\xi) = \left[f_Z(t) - \overline{F}_T(t)e^{t\xi} \left(f_C(t) - \xi\overline{F}_C(t)\right)\right]^b \left[\overline{F}_T(t)e^{t\xi} \left(f_C(t) - \xi\overline{F}_C(t)\right)\right]^{1-b}$$

and

$$h(t,b;\theta) = \left[ f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m,n}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m,n}(t|\overline{\phi}_m) \right) \right]^b \left[ \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m,n}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m,n}(t|\overline{\phi}_m) \right) \right]^{1-b}$$

with  $\overline{F}_{T,n}$ ,  $f_{Z,n}$ ,  $\overline{F}_{C,m,n}$  and  $f_{C,m,n}$  the estimators defined in the previous section. Let us introduce the following matrices

$$A\left(\theta\right) = \left(\mathbb{E}\left[\frac{\partial^2 \log h(Z,\delta;\theta)}{\partial \theta_i \partial \theta_j}\right]\right)_{1 \le i,j \le m+2} \text{ and } B\left(\theta\right) = \left(\mathbb{E}\left[\frac{\partial \log h(Z,\delta;\theta)}{\partial \theta_i}\frac{\partial \log h(Z,\delta;\theta)}{\partial \theta_j}\right]\right)_{1 \le i,j \le m+2}$$

We need the following regularity conditions to establish asymptotic properties of our estimators:

- $(\mathbf{H_1}):\ \Lambda_2'e^{-\Lambda_2}\ \text{and}\ \Lambda_1'e^{-\Lambda_1}\ \text{are element of}\ L^2(0,\infty).$
- (**H**<sub>2</sub>): The kernel K is bounded up by a positive constant  $K_{max}$ .
- (**H**<sub>3</sub>): I has unique minimum at  $\theta^*$ .
- (**H**<sub>4</sub>): (a)  $\theta^*$  is interior to  $\mathbb{R}^{m+2}_+$ .
  - (b)  $B(\theta^{\star})$  is non singular.
  - (c)  $\theta^{\star}$  is a regular point of  $A(\theta)$ .

**Remark 2.** Assumption  $(\mathbf{H_1})$  is essential for establishing the bounded expectation result in Proposition 5. Assumption  $(\mathbf{H_2})$  is a technical requirement ensuring that condition (5) of [60] is satisfied. Assumptions  $(\mathbf{H_3})$  and  $(\mathbf{H_4})$  serve as regularity conditions commonly adopted in maximum likelihood analysis to demonstrate asymptotic properties.

One requirement in the study of the asymptotic of the estimators is to ensure that the the random time Z admit finite moments for any order. This results is stated in the following proposition.

**Proposition 5.** Suppose that  $\Lambda_1$  and  $\Lambda_2$  satisfy  $(\mathbf{H_1})$ . Then, all moments of order k for Z exist, that is

$$\mathbb{E}(Z^{\kappa}) < \infty.$$

This allows us, in the next proposition, to show the asymptotic normality for the parameter vector  $\theta_n$ .

**Proposition 6.** Under assumptions  $(\mathbf{H_1}) - (\mathbf{H_4})$ ,  $\theta$  is identifiable and

$$\sqrt{n} \left(\theta_n - \theta\right) \xrightarrow[n \to +\infty]{\mathcal{L}} \mathcal{N} \left(0, C\left(\theta\right)\right)$$

where  $C(\theta) = A(\theta)^{-1} B(\theta) A(\theta)^{-1}$ .

The proof of Proposition 6 is done by verifying the conditions of Theorem 3.2 in [60]. In the next proposition, we finally establish the simple convergence for the joint distribution estimator and the Kendall's tau coefficient.

**Proposition 7.** Under assumptions  $(\mathbf{H_1}) - (\mathbf{H_4})$ ,

$$\lim_{n,m\to\infty}\widetilde{P}_{n,m}(t,s)=\widetilde{P}(t,s) \text{ and } \lim_{n\to\infty}\tau_n=\tau.$$

### 6 Simulation and real data analysis

#### 6.1 Simulation study

In this subsection, we evaluate the performance of the survival distribution estimators and the survival copula estimator. We considered N = 100 samples of sizes n = 30, 50, 100 under the following models:

- a)  $\Lambda_1(t) = 0.01t$  and  $\Lambda_2(t) = 0.02t$  defined on [0, 100], and  $(Y_t)_{0 \le t \le 100}$  is a Poisson process with intensity  $\lambda = 2$ .
- b)  $\Lambda_1(t) = -\ln(1 I_t(2.5, 6)) 0.4t$  and  $\Lambda_2(t) = -\ln(1 I_t(3, 2)) 0.4t$  defined on [0, 100], and  $(Y_t)_{0 \le t \le 100}$  is a Poisson process with intensity  $\lambda = 2$ , where  $I_t$  is the incomplete beta function.

The selection of the Bernstein polynomial degree m is determined by the available data based on the Akaike information criterion (AIC)

$$AIC(m) = 2(m+2) - 2\mathcal{L}_m(\xi, \phi_m).$$

The AIC is calculated to fit models across a grid of degree m, with the default upper bound set to n, so that the selected Bernstein degree minimizes the AIC. As a measure of performance, we evaluate the Mean Integrated Squared Error (MISE) of the mentioned estimators. Specifically, for a function F and its estimator  $\hat{F}$ , let

$$ISE(\widehat{F}) = \int_{a}^{b} \left[\widehat{F}(x) - F(x)\right]^{2} dx, \qquad (6.1)$$

where  $a = \min_{1 \le i \le n} (Z_i)$ ,  $b = \max_{1 \le i \le n} (Z_i)$ . Then for N sample of size n the MISE of  $\widehat{F}$  is given by the Monte Carlo approximation:

$$\mathrm{MISE}(\widehat{F}) = \frac{1}{N} \sum_{i=1}^{N} \mathrm{ISE}_{i}(\widehat{F}),$$

where  $\text{ISE}_i(\hat{F})$  denotes the value of the Integrated Squared Error (ISE) calculated on the *i*-th randomly generated sample.

Figures 1 and 2 display the survival estimators of  $\overline{F}_T$  and  $\overline{F}_C$ , respectively, under the two models a) and b). Figure 3 shows the estimation results of the functions  $\alpha$  and  $\beta$ , while Figure 4 displays the

results of the survival copula estimation. Figure 1 and 2 show that the survival estimators converge toward the true marginal survival functions. This is again a attractive feature of the proposed model since the dependency between T and C does not affect the KM estimator in this scenario. Figure 3 shows that the estimators of  $\alpha$  and  $\beta$  are constants under model a). This can be attributed to the fact that model a) is equivalent to a MO exponential distribution. As expected from the consistency result in Proposition 7, the discrepancy between  $\tilde{P}_{n,m}$  and  $\tilde{P}$  decreases, as indicated by the plots in Figure 4. We observe that, in all cases presented in Table 1, the value of the MISE decrease along with the sample size n, for appropriate choices of the degree m.



Figure 1: Kaplan Meier estimator of  $\overline{F}_T(t)$  with N = 100 iterations, for a size n = 100 for the model a) (left panel) and the model b) (right panel).



Figure 2: Bernstein estimators of  $\overline{F}_C(t)$  with N = 100 iterations, for a size n = 100 for the model a) (left panel) and the model b) (right panel).



Figure 3: Estimators of  $\alpha(t)$  and  $\beta(t)$  with N = 100 iterations, for a size n = 100 for the model a) (up) and the model b) (down).



Figure 4: Survival estimator of the joint law with N = 100 iterations, for a size n = 100 for the model a) (up) and the model b) (down).

	n	$\overline{F}_{T,n}$	$\overline{F}_{C,n,m}$	$\widetilde{P}_{n,m}$
Model a)	30	0.004518	0.015534	0.005218
	50	0.001618	0.001471	0.004331
	100	0.000278	0.000117	0.001298
Model b)	30	0.003696	0.003161	0.003198
	50	0.001418	0.000227	0.000573
	100	0.000180	0.000201	0.000289

Table 1: Results for MISE for N = 100 trials of the proposed estimators  $\overline{F}_{T,n}$ ,  $\overline{F}_{C,m,n}$  and  $\widetilde{P}_{n,m}$ , for n = 30, n = 50 and n = 100.

#### 6.2 Real data application

In this subsection, we apply the proposed estimation procedure to the joint retirement problem of married couples, using data from [34] and [44] drawn from eight waves of the Health and Retirement Study (every 2 years from 1992 to 2006). The duration variables (T, C) represent the retirement dates of the wife and the husband, respectively. The duration  $Z = \min(T, C)$  represents the first entry into retirement for the corresponding member of the household and  $\delta = \mathbb{1}_{\{T \leq C\}}$  indicates whether the wife retires first or the husband does. The latent stochastic processes  $(Y_t)_{t\geq 0}$  characterize the aging processes for the elderly. The sample consists of 821 households. We note that the supports are  $S_T = S_C = [1, 176]$ . Figure 5 depicts the KM survival estimator and the Bernstein estimator of  $\overline{F}_T$  and  $\overline{F}_C$ , respectively. Moreover, the results seem to contradict the assumption that the data come from the model proposed in [44]. In particular, with the estimation of the functions  $\alpha$  and  $\beta$ , Figure 6 shows that the results are not constant and do not always reach the expected values for [44]'s model, where  $\Lambda_1(t) = \Lambda_2(s) = 0$ . Figure 7 display the joint survival estimator of  $\widetilde{P}(t, s)$ .



Figure 5: Results for the survival distribution estimators of  $\overline{F}_T(t)$  and  $\overline{F}_C(t)$  with the real data.



Figure 6: Results for the estimators of  $\alpha(t)$  and  $\beta(t)$  with the real data.



Figure 7: Results for the estimators of  $\widetilde{P}(t,s)$  with the real data.

# 7 Conclusion

This paper presents a dependent censoring model for analyzing survival times, based on the canonical default model. We establish the identifiability of this model given censored data. Several theoretical results related to the model's functions, specifically the survival copula and cumulative hazard functions, are analyzed. A flexible non-/semi-parametric estimation method is proposed and illustrated through the analysis of synthetic and real datasets. This work can be considered a preliminary study for further investigations into the survival copula based on threshold-crossing models. Indeed, our study provides a theoretical foundation for characterizing the survival copula in the context of the canonical model, based on a latent non-homogeneous process.

# 8 Proofs

## Proof of Proposition 1:

Let  $t, s \ge 0$ . According to Theorem 2.2 in [32], we have

$$\widetilde{P}(t,s) = \mathbb{P}(T > t, C > s) = \overline{F}_T(t)\overline{F}_C(s)\frac{\psi_{Y_t+Y_s}(1)}{\psi_{Y_t}(1)\psi_{Y_s}(1)}.$$

Suppose that t < s. Using the fact that  $(Y_t)_{t \ge 0}$  has independent and stationary increments, and that

$$Y_t + Y_s = 2(Y_t - Y_0) + (Y_s - Y_t),$$

we obtain

$$\begin{split} \widetilde{P}(t,s) &= \overline{F}_T(t)\overline{F}_C(s)\frac{\psi_{2(Y_t-Y_0)+(Y_s-Y_t)}(1)}{\psi_{Y_t}(1)\psi_{Y_s}(1)} \\ &= \overline{F}_T(t)\overline{F}_C(s)\frac{\psi_{Y_t}(2)\psi_{Y_s-t}(1)}{\psi_{Y_t}(1)\psi_{Y_s}(1)} \\ &= \overline{F}_T(t)\overline{F}_C(s)\frac{e^{-t\kappa(2)}e^{-(s-t)\kappa(1)}}{e^{-t\kappa(1)}e^{-s\kappa(1)}} \\ &= \overline{F}_T(t)\overline{F}_C(s)e^{-t\kappa(2)}e^{-(s-t)\kappa(1)}e^{t\kappa(1)}e^{s\kappa(1)} \\ &= \overline{F}_T(t)\overline{F}_C(s)e^{-t\kappa(2)}e^{-s\kappa(1)+t\kappa(1)}e^{t\kappa(1)}e^{s\kappa(1)} \\ &= \overline{F}_T(t)\overline{F}_C(s)e^{t\xi} \\ &= \overline{F}_T(t)\overline{F}_C(s)e^{t\xi}. \end{split}$$

Similarly, in the case where s < t, we obtain

$$\widetilde{P}(t,s) = \overline{F}_T(t)\overline{F}_C(s)e^{s\xi}.$$

Therefore,

$$\widetilde{P}(t,s) = \overline{F}_T(t)\overline{F}_C(s)e^{\xi\min(t,s)} = \overline{F}_T(t)\overline{F}_C(s)\min(e^{\xi t}, e^{\xi s}).$$

In other hand, according to Theorem 2.2 in [32], we have

$$\overline{F}_T(t) = e^{-\Lambda_1(t) - \kappa(1)t}$$
 and  $\overline{F}_C(t) = e^{-\Lambda_2(t) - \kappa(1)t}$ .

Hence,

$$e^{\xi t} = \overline{F}_T(t)^{-\alpha(t)} = \overline{F}_C(t)^{-\beta(t)}, \quad \forall t \ge 0,$$

where

$$\alpha(t) = \frac{\xi t}{\Lambda_1(t) + t\kappa(1)}$$
 and  $\beta(t) = \frac{\xi t}{\Lambda_2(t) + t\kappa(1)}.$ 

It implies that

$$\widetilde{P}(t,s) = \widetilde{C}(\overline{F}_T(t), \overline{F}_C(s)) = \overline{F}_C(s)\overline{F}_T(t)\min\left(\overline{F}_T(t)^{-\alpha(t)}, \overline{F}_C(s)^{-\beta(s)}\right),$$

and therefore the survival copula is given by

$$\widetilde{\mathcal{C}}(u,v) = vu\min\left(u^{-\alpha\left(\overline{F}_{T}^{-}(u)\right)}, v^{-\beta\left(\overline{F}_{C}^{-}(v)\right)}\right).$$

## Proof of Proposition 2:

Let  $t \ge 0$ . Then, we have,

$$\begin{split} \overline{H}(t) &= \mathbb{P}(Z > t) \\ &= \mathbb{E}\left[\mathbb{P}(Z > t | Y_t)\right] \\ &= \mathbb{E}\left[\mathbb{P}\left(T > t, C > t | Y_t\right)\right] \\ &= \mathbb{E}\left[\mathbb{P}\left(\inf\{x \ge 0, e^{-M_{1,x}} \le U_1\} > t, \inf\{x \ge 0, e^{-M_{2,x}} \le U_2\} > t | Y_t\right)\right] \\ &= \mathbb{E}\left[\mathbb{P}\left(\bigcup_1 < e^{-M_{1,t}}, \bigcup_2 < e^{-M_{2,t}} | Y_t\right)\right] \\ &= \mathbb{E}\left[\mathbb{P}\left(U_1 < e^{-M_{1,t}}, |Y_t\right) \mathbb{P}\left(U_2 < e^{-M_{1,t}} | Y_t\right)\right] \\ &= \mathbb{E}\left[e^{-M_{1,t}} e^{-M_{2,t}}\right] \\ &= e^{-\Lambda_1(t)} e^{-\Lambda_2(t)} e^{-t\kappa(2)}. \end{split}$$

Let s < t, we have

$$\mathbb{P}(T > t, C > s) = \overline{F}_T(t)\overline{F}_C(s)e^{s\xi}.$$
(8.1)

Therefore, according to Theorem 1 of [7], we obtain

$$f_{Z,\delta=0}(t) = -\lim_{y \to t} \frac{\partial \mathbb{P}(T > t, C > y)}{\partial y}.$$

Using Equation (8.1), we obtain

$$\frac{\partial \mathbb{P}(T > t, C > y)}{\partial y} = \frac{\partial \left[\overline{F}_T(t)\overline{F}_C(y)e^{y\xi}\right]}{\partial y} = \overline{F}_T(t) \left[-f_C(y)e^{y\xi} + \xi \overline{F}_C(y)e^{y\xi}\right].$$

Hence,

$$\begin{split} f_{Z,\delta=0}(t) &= -\lim_{y \to t} \overline{F}_T(t) \left[ -f_C(y) e^{y\xi} + \xi \overline{F}_C(y) e^{y\xi} \right] \\ &= -\overline{F}_T(t) \left[ \overline{F}'_C(t) e^{t\xi} + \xi \overline{F}_C(t) e^{t\xi} \right] \\ &= \overline{F}_T(t) e^{t\xi} \left[ -\overline{F}'_C(t) - \xi \overline{F}_C(t) \right] \\ &= \overline{F}_T(t) e^{t\xi} \left[ (\Lambda'_2(t) + \kappa(1)) \overline{F}_C(t) - \xi \overline{F}_C(t) \right] \\ &= \overline{F}_C(t) \overline{F}_T(t) e^{t\xi} \left[ \Lambda'_2(t) + \kappa(1) - \xi \right] \\ &= \overline{H}(t) [\Lambda'_2(t) + \kappa(1) - \xi]. \end{split}$$

Moreover,

$$f_{Z}(t) = -\overline{H}'(t)$$
  
=  $\left[\Lambda_{1}'(t) + \Lambda_{2}'(t) + \kappa(2)\right] e^{-\Lambda_{1}(t) - \Lambda_{2}(t) - t\kappa(2)}$   
=  $\left[\Lambda_{1}'(t) + \Lambda_{2}'(t) + \kappa(2)\right] \overline{H}(t).$ 

Consequently,

$$\begin{aligned} f_{Z,\delta=1}(t) &= f_Z(t) - f_{Z,\delta=0}(t) \\ &= \left[\Lambda'_1(t) + \Lambda'_2(t) + \kappa(2)\right] \overline{H}(t) - \overline{H}(t) [\Lambda'_2(t) + \kappa(1) - \xi] \\ &= \overline{H}(t) \left[\Lambda'_1(t) + \Lambda'_2(t) + \kappa(2) - \Lambda'_2(t) - \kappa(1) + 2\kappa(1) - \kappa(2)\right] \\ &= \overline{H}(t) \left[\Lambda'_1(t) + \kappa(1)\right], \end{aligned}$$

which implies that

$$\frac{f_{Z,\delta=1}(t)}{\overline{H}(t)} = \Lambda'_1(t) + \kappa(1).$$

On the other hand, we have

$$\frac{f_T(t)}{\overline{F}_T(t)} = \frac{-\overline{F}_T(t)}{\overline{F}_T(t)} \\
= \frac{(\Lambda_1'(t) + \kappa(1))e^{-\Lambda_1(t) - t\kappa(1)}}{e^{-\Lambda_1(t) - t\kappa(1)}} \\
= \Lambda_1'(t) + \kappa(1).$$

Then

$$\frac{f_{Z,\delta=1}(t)}{\overline{H}(t)} = \frac{f_T(t)}{\overline{F}_T(t)}.$$

Similarly, we have

$$\frac{f_{Z,\delta=0}(t)}{\overline{H}(t)} = \frac{f_C(t)}{\overline{F}_C(t)} - \xi.$$

#### **Proof of Proposition 3:**

Consider the model defined in (1.1)-(1.2). Let  $X_1, X_2$ , and  $X_3$  be three independent random variables with cumulative Hazard functions  $\Lambda_{X_1}, \Lambda_{X_2}$ , and  $\Lambda_{X_3}$ , respectively. Suppose that, for  $t \ge 0$ 

$$\begin{cases} \Lambda_{X_1}(t) = \Lambda_1(t) + t\kappa(1) - \xi t\\ \Lambda_{X_2}(t) = \Lambda_2(t) + t\kappa(1) - \xi t\\ \Lambda_{X_3}(t) = \xi t. \end{cases}$$

$$(8.2)$$

Define the random variables X and Y as follows

$$X = \min(X_1, X_3),$$
 and  $Y = \min(X_2, X_3).$ 

Then, according to Proposition 1 in [27], the survival copula of (X, Y) is given by

$$\tilde{C}_{(X,Y)}(u,v) = uv \min\left(u^{-\alpha_1(\overline{F}_X^-(u))}, v^{-\alpha_2(\overline{F}_Y^-(v))}\right),$$

where

$$\alpha_1(t) = \frac{\Lambda_{X_3}(t)}{\Lambda_{X_3}(t) + \Lambda_{X_1}(t)} = \frac{\xi t}{\Lambda_1(t) + t\kappa(1)},$$

and

$$\alpha_2(t) = \frac{\Lambda_{X_3}(t)}{\Lambda_{X_3}(t) + \Lambda_{X_2}(t)} = \frac{\xi t}{\Lambda_2(t) + t\kappa(1)}$$

Therefore (X, Y) has the same copula as (T, C). In the other hand,

$$\overline{F}_X(t) = \overline{F}_{X_1}(t)\overline{F}_{X_3}(t) = e^{-\Lambda_1(t) - \Lambda_3(t)} = e^{-\Lambda_1(t) - t\kappa(1)} = \overline{F}_T(t).$$

Similarly, we have

$$\overline{F}_Y(t) = \overline{F}_{X_2}(t)\overline{F}_{X_3}(t) = e^{-\Lambda_2(t) - \Lambda_3(t)} = e^{-\Lambda_2(t) - t\kappa(1)} = \overline{F}_C(t).$$

Then, (X, Y) has the same distribution as (T, C). Moreover,

 $T \stackrel{d}{=} \min(X_1, X_3)$  and  $C \stackrel{d}{=} \min(X_2, X_3)$ .

#### **Proof of Proposition 4:**

 $e^{-}$ 

Suppose that

 $f_{Z,\delta=1,\Gamma^{1}}(t) = f_{Z,\delta=1,\Gamma^{2}}(t) \quad \text{and} \quad f_{Z,\delta=0,\Gamma^{1}}(t) = f_{Z,\delta=1,\Gamma^{2}}(t),$ 

where  $\Gamma^i = (\Lambda_1^i, \Lambda_2^i, \kappa(1)^i, \kappa(2)^i)$  for i = 1, 2. Then, for  $t \in [0, \infty[$ , we have

$$-\Lambda_1^1(t) - \Lambda_2^1(t) - t\kappa(2)^1 (\Lambda_1^1(t)' + \kappa(1)^1) = e^{-\Lambda_1^2(t) - \Lambda_2^2(t) - t\kappa(2)^2} (\Lambda_1^2(t)' + \kappa(1)^2),$$
(8.3)

and

$$e^{-\Lambda_1^1(t) - \Lambda_2^1(t) - t\kappa(2)^1} \left( \Lambda_2^1(t)' + \kappa(2)^1 - \kappa(1)^1 \right)$$

$$= e^{-\Lambda_1^2(t) - \Lambda_2^2(t) - t\kappa(2)^2} \left( \Lambda_2^2(t)' + \kappa(2)^2 - \kappa(1)^2 \right).$$
(8.4)

Since  $\Lambda'_1(0) = \Lambda'_2(0) = 0$ , then for t = 0, Equation (8.3) implies that  $\kappa(1)^1 = \kappa(1)^2$ . Moreover, for t = 0, Equation (8.4) implies that  $\kappa(2)^1 = \kappa(2)^2$ . By summing Equations (8.3) and (8.4), we obtain

$$e^{-E^{1}(t)}(E^{1}(t)') = e^{-E^{2}(t)}(E^{2}(t)'),$$

where  $E^{i}(t) = \Lambda_{1}^{i}(t) + \Lambda_{2}^{i}(t) + \kappa(2)^{i}$ . Then  $\left[e^{-E^{1}(t)}\right]' = \left[e^{-E^{2}(t)}\right]'$ . It implied that, for  $x \in [0, \infty[$ ,  $\int_{0}^{x} \left[e^{-E^{1}(t)}\right]' dt = \int_{0}^{x} \left[e^{-E^{2}(t)}\right]' dt$ ,

and therefore  $e^{-E^1(x)} = e^{-E^2(x)}$ , since  $\Lambda_1^1(0) = \Lambda_1^2(0) = \Lambda_2^1(0) = \Lambda_2^2(0) = 0$ . Substituting this result into Equation (8.3), we obtain  $\Lambda_1^1(t)' = \Lambda_1^2(t)'$ . Then

$$\int_0^x \Lambda_1^1(t)' dt = \int_0^x \Lambda_1^2(t)' dt,$$

and therefore  $\Lambda_1^1(x) = \Lambda_1^2(x)$ , since  $\Lambda_1^1(0) = \Lambda_1^2(0) = 0$ . Substituting this result into Equation (8.4), we obtain  $\Lambda_2^1(t)' = \Lambda_2^2(t)'$ . Then

$$\int_0^x \Lambda_2^1(t)' dt = \int_0^x \Lambda_2^2(t)' dt$$

and therefore  $\Lambda_2^1(x) = \Lambda_2^2(x)$ , since  $\Lambda_2^1(0) = \Lambda_2^2(0) = 0$ . Finally, we conclude that  $\Gamma^1 = \Gamma^2$ .

### **Proof of Proposition 5:**

Suppose that  $\Lambda_1$  and  $\Lambda_2$  satisfy  $(\mathbf{H_1})$ . Let  $k \in \mathbb{N}$ . We have

$$\begin{split} \mathbb{E}(Z^k) &= \int_0^\infty t^k \left(\Lambda_1'(t) + \Lambda_2'(t) + \kappa(2)\right) e^{-\Lambda_1(t) - \Lambda_2(t) - t\kappa(2)} dt \\ &= \int_0^\infty t^k \left(\Lambda_1'(t) + \Lambda_2'(t) + \kappa(2)\right) e^{-\Lambda_1(t) - \Lambda_2(t) - t\frac{\kappa(2)}{2} - t\frac{\kappa(2)}{2}} dt \\ &\leq \left(\int_0^\infty t^{2k} e^{-t\kappa(2)} dt\right)^{1/2} \times \left(\int_0^\infty \left(\Lambda_1'(t) + \Lambda_2'(t) + \kappa(2)\right)^2 e^{-2\Lambda_1(t) - 2\Lambda_2(t) - t\kappa(2)} dt\right)^{1/2} \\ &= \left(\frac{(2k)!}{\kappa(2)^{2k+1}}\right)^{1/2} \times I^{1/2}, \end{split}$$

where

$$\begin{split} I &= \int_{0}^{\infty} \left( \Lambda_{1}^{'}(t) + \Lambda_{2}^{'}(t) + \kappa(2) \right)^{2} e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2)} dt \\ &= \int_{0}^{\infty} \Lambda_{1}^{'}(t)^{2} e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2)} dt + \int_{0}^{\infty} \Lambda_{2}^{'}(t)^{2} e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - 2t\kappa(2)} dt \\ &+ \kappa(2)^{2} \int_{0}^{\infty} e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2)} dt + 2\int_{0}^{\infty} \Lambda_{1}^{'}(t)\Lambda_{2}^{'}(t) e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2)} dt \\ &+ 2\kappa(2) \int_{0}^{\infty} \Lambda_{1}^{'}(t) e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2)} dt + 2\kappa(2) \int_{0}^{\infty} \Lambda_{2}^{'}(t) e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2)} dt \\ &\leq \int_{0}^{\infty} \Lambda_{1}^{'}(t)^{2} e^{-2\Lambda_{1}(t)} dt + \int_{0}^{\infty} \Lambda_{2}^{'}(t)^{2} e^{-2\Lambda_{2}(t)} dt + \kappa(2)^{2} \int_{0}^{\infty} e^{-t\kappa(2)} dt \\ &+ 2\int_{0}^{\infty} \Lambda_{1}^{'}(t)\Lambda_{2}^{'}(t) e^{-\Lambda_{1}(t) - \Lambda_{2}(t)} dt + 2\kappa(2) \int_{0}^{\infty} \Lambda_{1}^{'}(t) e^{-\Lambda_{1}(t)} dt + 2\kappa(2) \int_{0}^{\infty} \Lambda_{2}^{'}(t) e^{-\Lambda_{2}(t)} dt \\ &= \int_{0}^{\infty} \Lambda_{1}^{'}(t)^{2} e^{-2\Lambda_{1}(t)} dt + \int_{0}^{\infty} \Lambda_{2}^{'}(t)^{2} e^{-2\Lambda_{2}(t)} dt + 2\int_{0}^{\infty} \Lambda_{1}^{'}(t)\Lambda_{2}^{'}(t) e^{-\Lambda_{1}(t) - \Lambda_{2}(t)} dt + 5\kappa(2). \end{split}$$

Using Hölder's inequality we obtain

$$\begin{split} \int_0^\infty \Lambda_2'(t)\Lambda_1'(t)e^{-\Lambda_1(t)-\Lambda_2(t)}dt &\leq \left(\int_0^\infty \Lambda_1'(t)^2 e^{-2\Lambda_1(t)}dt\right)^{1/2} \times \left(\int_0^\infty \Lambda_2'(t)^2 e^{-2\Lambda_2(t)}dt\right)^{1/2} \\ &< \infty, \end{split}$$

which implies that I exists and then  $\mathbb{E}(Z^k) < \infty$ .

#### Proof of Proposition 6:

In order to prove this proposition, we propose to verify the conditions (1) - (6) of [60].

The independent random vector  $(Z_i, \delta_i)$  for  $i = 1 \dots n$  have a common joint distribution G on  $[0, 1]^2$  with measurable Radon-Nikodym density given by

$$g(t,b;\xi) = f_{Z,\delta=b}(t)$$
  
=  $[f_{Z,\delta=1}(t)]^{b} [f_{Z,\delta=0}(t)]^{1-b}$   
=  $\left[f_{Z}(t) - \overline{F}_{T}(t)e^{t\xi} \left(f_{C}(t) - \xi \overline{F}_{C}(t)\right)\right]^{b} \left[\overline{F}_{T}(t)e^{t\xi} \left(f_{C}(t) - \xi \overline{F}_{C}(t)\right)\right]^{1-b}.$ 

The Radon-Nykodym density function h defined by

$$h(t,b;\theta) = h(t,b;\xi,\bar{\phi}_m)$$
  
=  $\left[f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left(f_{C,m}(t|\bar{\phi}_m) - \xi\overline{F}_{C,m}(t|\bar{\phi}_m)\right)\right]^b$   
 $\times \left[\overline{F}_{T,n}(t)e^{t\xi} \left(f_{C,m}(t|\bar{\phi}_m) - \xi\overline{F}_{C,m}(t|\bar{\phi}_m)\right)\right]^{1-b},$ 

is measurable on (t, b) for every  $\theta \in \mathbb{R}^{m+2}_+$  and continuous on  $\theta$  of every  $(t, b) \in [0, 1]^2$ . Then conditions (1) and (2) of [60] are verified.

In order to show condition (3)a) of [60], we need first to show that  $\mathbb{E}(\log g(Z, \delta; \xi))$  exist.

\* Suppose that  $\delta = 0$ . Based on assumption  $(\mathbf{H}_1)$ , we have  $\Lambda'_2(t)e^{-\Lambda_2(t)}$  and  $\Lambda'_1(t)e^{-\Lambda_1(t)}$  are element of  $L^2(0,\infty)$ . Since  $\overline{F}_T(Z) \leq 1$  and  $f_C(Z) - \xi \overline{F}_C(Z) \leq f_C(Z)$ , then

$$\mathbb{E} \left( \log g(Z, \delta; \xi) \right) = \mathbb{E} \left[ \log \left( \overline{F}_T(Z) e^{Z\xi} \left( f_C(Z) - \xi \overline{F}_C(Z) \right) \right) \right] \\ \leq \mathbb{E} \left[ \log \left( f_C(Z) \right) \right] + \mathbb{E} (\log (e^{Z\xi})) \\ \leq \mathbb{E} \left( f_C(Z) \right) + \xi \mathbb{E}(Z) \\ \leq \int_0^\infty f_C(t) f_Z(t) dt + \xi \mathbb{E}(Z) \\ = \int_0^\infty \left( \Lambda'_2(t) + \kappa(1) \right) \left( \Lambda'_1(t) + \Lambda'_2(t) + \kappa(2) \right) e^{-\Lambda_1(t) - 2\Lambda_2(t) - t\kappa(2) - t\kappa(1)} dt \\ + \xi \mathbb{E}(Z) \\ = I_1 + I_2 + (\kappa(2) + \kappa(1)) I_3 + \kappa(1) I_4 + \kappa(1) \kappa(2) I_5 + \xi \mathbb{E}(Z)$$

where,

$$\begin{split} I_{1} &= \int_{0}^{\infty} \Lambda_{2}'(t) \Lambda_{1}'(t) e^{-\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2) - t\kappa(1)} dt \\ I_{2} &= \int_{0}^{\infty} \Lambda_{2}'(t)^{2} e^{-\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2) - t\kappa(1)} dt \\ I_{3} &= \int_{0}^{\infty} \Lambda_{2}'(t) e^{-\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2) - t\kappa(1)} dt \\ I_{4} &= \int_{0}^{\infty} \Lambda_{1}'(t) e^{-\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2) - t\kappa(1)} dt \\ I_{5} &= \int_{0}^{\infty} e^{-\Lambda_{1}(t) - 2\Lambda_{2}(t) - t\kappa(2) - t\kappa(1)} dt. \end{split}$$

In other hand, from Holder inequality we have

$$\begin{split} I_{1} &\leq \left[ \int_{0}^{\infty} \Lambda_{2}'(t)^{2} e^{-4\Lambda_{2}(t) - 2\kappa(2)t - 2\kappa(1)t} dt \right]^{1/2} \\ &\times \left[ \int_{0}^{\infty} \Lambda_{1}'(t)^{2} e^{-2\Lambda_{1}(t)} dt \right]^{1/2} \\ &\leq \left[ \int_{0}^{\infty} \Lambda_{2}'(t)^{2} e^{-2\Lambda_{2}(t)dt} \right]^{1/2} \times \left[ \int_{0}^{\infty} \Lambda_{1}'(t)^{2} e^{-2\Lambda_{1}(t)} dt \right]^{1/2} < \infty, \\ I_{2} &\leq \int_{0}^{\infty} \Lambda_{2}'(t)^{2} e^{-2\Lambda_{2}(t)} dt < \infty, \\ I_{3} &\leq \int_{0}^{\infty} \Lambda_{2}'(t) e^{-\Lambda_{2}(t)} dt \\ &= -\int_{0}^{\infty} \left( e^{-\Lambda_{2}(t)} \right)' dt \\ &= 1. \end{split}$$

Similarly to  $I_3$ , we have  $I_4 < \infty$ . In addition, we have

$$I_5 \le \int_0^\infty e^{-t(\kappa(2)+\kappa(1))} dt$$
$$= \frac{1}{\kappa(2)+\kappa(1)} < \infty.$$

Based on Theorem 5,  $\mathbb{E}(Z) < \infty$ , we conclude that  $\mathbb{E}(\log g(Z, \delta; \xi)) < \infty$ .

 $\star\,$  Now, suppose that  $\delta=1,$  then

$$\begin{split} \mathbb{E} \left( \log g(Z, \delta; \xi) \right) \\ &= \mathbb{E} \left[ \log \left( f_Z(Z) - \overline{F}_T(Z) e^{Z\xi} \left( f_C(Z) - \xi \overline{F}_C(Z) \right) \right) \right] \\ &\leq \mathbb{E} \left[ \log \left( f_Z(Z) + \overline{F}_T(Z) e^{Z\xi} \xi \overline{F}_C(Z) \right) \right] \\ &\leq \mathbb{E} \left[ \log \left( f_Z(Z) + \xi e^{Z\xi} \right) \right] \\ &\leq \mathbb{E} \left( f_Z(Z) + \xi e^{Z\xi} \right) \\ &= \mathbb{E} \left( f_Z(Z) + \xi e^{Z\xi} \right) \\ &= \int_0^\infty \left( \Lambda_1'(t) + \Lambda_2'(t) + \kappa(2) \right)^2 e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt \\ &+ \xi \mathbb{E} (e^{Z\xi}) \\ &= \int_0^\infty \Lambda_1'(t)^2 e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt + \int_0^\infty \Lambda_2'(t)^2 e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt \\ &+ \kappa(2)^2 \int_0^\infty e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt + 2\int_0^\infty \Lambda_1'(t) \Lambda_2'(t) e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt \\ &+ 2\kappa(2) \int_0^\infty \Lambda_1'(t) e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt + 2\kappa(2) \int_0^\infty \Lambda_2'(t) e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2t\kappa(2)} dt \\ &+ \xi \mathbb{E} (e^{Z\xi}) \\ &= J_1 + J_2 + \kappa(2)^2 J_3 + 2J_4 + 2\kappa(2) J_5 + 2\kappa(2) J_6 + \xi J_7. \end{split}$$

We have

$$J_1 = \int_0^\infty \Lambda'_1(t)^2 e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2\kappa(2)t} dt$$
  
$$\leq \int_0^\infty \Lambda'_1(t)^2 e^{-2\Lambda_1(t)} dt < \infty.$$

Similarly to  $J_1$ , we have  $J_2 < \infty$ .

$$J_3 = \int_0^\infty e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2\kappa(2)t} dt$$
  
$$\leq \int_0^\infty e^{-2\kappa(2)t} dt$$
  
$$= \frac{1}{2\kappa(2)} < \infty.$$

Based on Hölder's inequality, we have

$$J_{4} = \int_{0}^{\infty} \Lambda_{1}'(t) \Lambda_{2}'(t) e^{-2\Lambda_{1}(t) - 2\Lambda_{2}(t) - 2\kappa(2)t} dt$$
  

$$\leq \int_{0}^{\infty} \Lambda_{1}'(t) \Lambda_{2}'(t) e^{-\Lambda_{1}(t) - \Lambda_{2}(t)} dt$$
  

$$\leq \left[ \int_{0}^{\infty} \Lambda_{1}'(t)^{2} e^{-2\Lambda_{1}(t)} dt \right]^{1/2} \left[ \int_{0}^{\infty} \Lambda_{2}'(t)^{2} e^{-2\Lambda_{2}(t)} dt \right]^{1/2} < \infty,$$

and

$$J_5 = \int_0^\infty \Lambda'_1(t) e^{-2\Lambda_1(t) - 2\Lambda_2(t) - 2\kappa(2)t} dt$$
  
$$\leq \int_0^\infty \Lambda'_1(t) e^{-\Lambda_1(t)} dt$$
  
$$= -\int_0^\infty \left( e^{-\Lambda_1(t)} \right)' dt$$
  
$$= 1.$$

Similarly to  $J_5$ , we have  $J_6 < 1$ . In addition, we have

$$\begin{aligned} J_{7} &= \mathbb{E}(e^{Z\xi}) \\ &= \int_{0}^{\infty} \left( \Lambda_{1}'(t) + \Lambda_{2}'(t) + \kappa(2) \right) e^{-\Lambda_{1}(t) - \Lambda_{2}(t) - t\kappa(2) + t\xi} dt \\ &= \int_{0}^{\infty} \left( \Lambda_{1}'(t) + \Lambda_{2}'(t) + \kappa(2) \right) e^{-\Lambda_{1}(t) - \Lambda_{2}(t) - 2t(\kappa(2) - \kappa(1))} dt \\ &\leq \int_{0}^{\infty} \Lambda_{1}'(t) e^{-\Lambda_{1}(t)} dt + \int_{0}^{\infty} \Lambda_{2}'(t) e^{-\Lambda_{2}(t)} dt + \kappa(2) \int_{0}^{\infty} e^{-2t(\kappa(2) - \kappa(1))} dt \\ &= -\int_{0}^{\infty} \left( e^{-\Lambda_{1}(t)} \right)' dt - \int_{0}^{\infty} \left( e^{-\Lambda_{2}(t)} \right)' dt + \frac{\kappa(2)}{2(\kappa(2) - \kappa(1))} \\ &= 2 + \frac{\kappa(2)}{2(\kappa(2) - \kappa(1))} < \infty. \end{aligned}$$

We conclude that  $\mathbb{E}(\log g(Z, \delta; \xi)) < \infty$ .

Second, we need to show that  $|\log(h(t, b; \theta))| \le m(t, b)$ , where is integrable with respect to g, for all  $\theta$ . To do this, let  $t \in [0, 1]$ . Assume that, there exist k > 0, such that

$$\kappa(2) - \kappa(1) \ge k. \tag{8.5}$$

Since  $\Lambda_1$  and  $\Lambda_2$  are continuous differentiable functions, then there exist  $\rho > 0$  such that

$$\Lambda_1(t) \le \rho, \quad \text{and} \quad \Lambda_2(t) \le \rho, \quad 0 \le t \le 1.$$
(8.6)

• If b = 0, then we have

$$\left|\log\left(h(t,b;\theta)\right)\right| = \left|\log\left(\overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)\right)\right|.$$

**Case 1:** If  $\overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \leq 1$ , then we have

$$\log (h(t,b;\theta))| = \left| \log \left( \overline{F}_{T,n}(t) e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \right) \right|$$
$$= -\log \left( \overline{F}_{T,n}(t) e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \right)$$

In other hand, we have

$$\overline{F}_T(t)e^{t\xi} \left( f_C(t) - \xi \overline{F}_C(t) \right) = (\Lambda_2'(t) + \kappa(2) - \kappa(1))e^{-\Lambda_1(t) - \Lambda_2(t) - \kappa(2)t}$$
$$\geq (\Lambda_2'(t) + \kappa(2) - \kappa(1))e^{-2\rho - \kappa(3)}$$
$$\geq (\kappa(2) - \kappa(1))e^{-2\rho - \kappa(3)}$$
$$\geq ke^{-2\rho - \kappa(3)}.$$

Thus,

$$-\log\left(\overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)\right) \leq -\log(k)$$

which is integrable with respect to the density g.

Case 2: If  $\overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right) \ge 1$ , then we have  $|\log(h(t,b;\theta))| = \left|\log\left(\overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)\right)\right|$   $= \log\left(\overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)\right)$   $\le \log\left(\overline{F}_{T,n}(t)e^{t\xi}f_{C,m}(t|\overline{\phi}_m)\right)$   $\le \log(f_{C,m}(t|\overline{\phi}_m))$  $\le \log(\kappa(3)).$ 

In fact,

$$f_C(t) = \left(\Lambda_2(t)' + \kappa(1)\right) e^{-\Lambda_2(t) - \kappa(1)t}$$
  
$$\leq \kappa(1) e^{-\Lambda_2(t) - \kappa(1)t}$$
  
$$\leq \kappa(3),$$

which is integrable with respect to the density g.

• If b = 1, we obtain

$$\left|\log\left(h(t,b;\theta)\right)\right| = \left|\log\left(f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_{m}) - \xi\overline{F}_{C,m}(t|\overline{\phi}_{m})\right)\right)\right|$$
  
• If  $f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_{m}) - \xi\overline{F}_{C,m}(t|\overline{\phi}_{m})\right) \le 1$  then

**Case 1:** If 
$$f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \leq 1$$
, then  
 $\left| \log \left( h(t,b;\theta) \right) \right| = -\log \left( f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \right)$   
 $\leq -\log \left( f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} f_{C,m}(t|\overline{\phi}_m) \right).$ 

In other hand, we have

$$f_{Z}(t) - \overline{F}_{T}(t)e^{t\xi}f_{C}(t) = \left(\Lambda_{1}(t)' + \Lambda_{2}(t)' + \kappa(2)\right)e^{-\Lambda_{1}(t) - \Lambda_{2}(t) - \kappa(2)t} - (\Lambda_{2}(t)' + \kappa(1))e^{-\Lambda_{1}(t) - \Lambda_{2}(t) - \kappa(2)t} = \left(\Lambda_{1}(t)' + \kappa(2) - \kappa(1)\right)e^{-\Lambda_{1}(t) - \Lambda_{2}(t) - \kappa(2)t} \geq (\kappa(2) - \kappa(1))e^{-2\rho - \kappa(3)} \geq ke^{-2\rho - \kappa(3)}.$$
(8.7)

Then

$$f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi}f_{C,m}(t|\overline{\phi}_m) \ge ke^{-2\rho-\kappa(3)},$$

and

$$\left|\log(h(t,b;\theta))\right| \le -\log\left(ke^{-2\rho-\kappa(3)}\right),$$

which is integrable with respect to the density g.

**Case 2:** If 
$$f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \ge 1$$
, then  
 $\left| \log \left( h(t,b;\theta) \right) \right| = \log \left( f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \right)$   
 $\le \log \left( f_{Z,n}(t) \right).$ 

In other hand, we have

$$f_Z(t) = \left(\Lambda_1(t)' + \Lambda_2(t)' + \kappa(2)\right) e^{-\Lambda_1(t) - \Lambda_2(t) - \kappa(2)t}$$
  
$$\leq \kappa(2) \leq \kappa(3).$$

Then,

$$\left|\log\left(h(t,b;\theta)\right)\right| \le \log(\kappa(3)),$$

which is integrable with respect to the density g.

Then, condition (3)a) is satisfied. Based on Assumption  $(\mathbf{H}_3)$ , I has unique minimum at  $\theta^*$ , and then condition (3)b) is verified.

In order to prove condition (4), we note first that,

$$\log(h(t,b;\theta)) = b \log \left[ f_{Z,n}(t) - \overline{F}_{T,n}(t) e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \right] \\ \times (1-b) \log \left[ \overline{F}_{T,n}(t) e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \right].$$

$$h_{1}(t,b;\theta) = \frac{\partial \log(h(t,b;\theta))}{\partial \xi}$$

$$= -\frac{b\overline{F}_{T,n}(t) \left[\xi e^{t\xi} f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(e^{t\xi} + \xi^{2}e^{t\xi})\right]}{f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left(f_{C,m}(t|\overline{\phi}_{m}) - \xi\overline{F}_{C,m}(t|\overline{\phi}_{m})\right)}$$

$$+ \frac{(1-b) \left[\xi f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2})\right]}{\left(f_{C,m}(t|\overline{\phi}_{m}) - \xi\overline{F}_{C,m}(t|\overline{\phi}_{m})\right)}.$$

For  $k = 0, \ldots, m$ 

$$h_{2,k}(t,b;\theta) = \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k}$$
  
=  $-\frac{b\overline{F}_{T,n}(t)e^{t\xi} (R_m(t)-\xi)}{f_{Z,n}(t)-\overline{F}_{T,n}(t)e^{t\xi} (f_{C,m}(t|\overline{\phi}_m)-\xi\overline{F}_{C,m}(t|\overline{\phi}_m))}$   
+  $\frac{(1-b) (R_m(t)-\xi)}{(f_{C,m}(t|\overline{\phi}_m)-\xi\overline{F}_{C,m}(t|\overline{\phi}_m))},$ 

where  $R_m(t) = \sum_{k=0}^{m} b'_{k,m}(t)$ . Then, the functions  $h_1$  and  $h_{2,k}$  for  $k = 0, \ldots m$  are measurable functions of (t, b) for each  $\theta \in \mathbb{R}^{m+2}$  and continuously differentiable functions of  $\theta$  for each  $(t, b) \in \mathbb{R}_+ \times [0, 1]$ . Then condition (4) is satisfied.

In order to show condition (5), we note that

• If  $f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \ge 1$  and  $f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \ge 1$ , then

$$\frac{1}{f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right)} \le 1 \text{ and } \frac{1}{f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m)} \le 1.$$

• If  $f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right) \leq 1$  and  $f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \leq 1$ , then based on Equations (8.5), (8.6) and (8.7), we have

$$\frac{1}{f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi} \left( f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right)} \le \frac{1}{ke^{-2\rho - \kappa(3)}}$$

Also, we have

$$\frac{1}{f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m)} \le \frac{1}{ke^{\rho - \kappa(3)}}.$$

In fact

$$f_C(t) - \xi \overline{F}_C(t) = e^{-\Lambda_2(t) - t\kappa(1)} (\Lambda'_2(t) + \kappa(1) - \xi)$$
  
 
$$\geq k e^{-\rho - \kappa(3)}.$$

Then, in general, there exist a constants  $\mu, \nu > 0$  such that,

$$\frac{1}{f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)} \le \mu \text{ and } \frac{1}{f_{C,m}(t|\overline{\phi}_m) - \xi\overline{F}_{C,m}(t|\overline{\phi}_m)} \le \nu.$$

$$\begin{split} U_{1}(t,b;\theta) &= \left| \frac{\partial^{2} \log(h(t,b;\theta))}{\partial^{2}\xi} \right| \\ &= \left| \frac{\partial h_{1}(t,b;\theta)}{\partial \xi} \right| \\ &\leq \mu^{2} \left| e^{t\xi} f_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2}) - e^{t\xi} \overline{F}_{C,m}(t|\overline{\phi}_{m})(3\xi+\xi^{3}) \right| \\ &\times \left| f_{Z,n}(t) - e^{t\xi} \overline{F}_{T,n}(t) f_{C,m}(t|\overline{\phi}_{m}) + \xi e^{t\xi} \overline{F}_{T,n}(t) \overline{F}_{C,m}(t|\overline{\phi}_{m}) \right| \\ &+ \mu^{2} \left| \xi e^{t\xi} f_{C,m}(t|\overline{\phi}_{m}) - e^{t\xi} \overline{F}_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2}) \right| \\ &\times \left| \overline{F}_{T,n}(t) e^{t\xi} (\xi f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2})) \right| \\ &+ \nu^{2} \left| \left[ f_{C,m}(t|\overline{\phi}_{m}) - 2\xi \overline{F}_{C,m}(t|\overline{\phi}_{m}) \right] \left[ f_{C,m}(t|\overline{\phi}_{m}) - \xi \overline{F}_{C,m}(t|\overline{\phi}_{m}) \right] \right| \\ &+ \nu^{2} \left| \overline{F}_{C,m}(t|\overline{\phi}_{m})(\xi f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2})) \right| \\ &\leq \mu^{2} \left[ e^{t\xi} R_{m}(t)(1+\xi^{2}) + e^{t\xi}(3\xi+\xi^{3}) \right] \times \left[ \frac{K_{max}}{h} + e^{t\xi} R_{m}(t) + \xi e^{t\xi} \right] \\ &+ \nu^{2} \left( \left[ \xi e^{t\xi} R_{m}(t) + e^{t\xi}(1+\xi^{2}) \right]^{2} + \left[ R_{m}(t) + 2\xi \right] \times \left[ R_{m}(t) + \xi \right] + \left[ \xi R_{m}(t) + 1 + \xi \right] \right) := S_{1}(t). \end{split}$$

Since the function  $S_1$  is continuous on the compact interval [0, 1], then it has a maximum  $\psi_1$ . Then

$$U_1(t,b;\theta) \leq \psi_1.$$

Then  $U_1$  is dominated by an integrable function with respect to the density g for all  $(t, b) \in [0, 1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(ii) For k = 0, ..., m,

$$\begin{aligned} U_{2,k}(t,b;\theta) &= \left| \frac{\partial^2 \log(h(t,b;\theta))}{\partial^2 \phi_k} \right| \\ &= \left| \frac{\partial h_{2,k}(t,b;\theta)}{\partial \phi_k} \right| \\ &\leq \left| \frac{b \left[ \overline{F}_{T,n}(t) e^{t\xi} (R_m(t) - \xi) \right]^2}{\left[ f_{Z,n}(t) - \overline{F}_{T,n}(t) e^{t\xi} (f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m)) \right]^2} \right| \\ &+ \left| \frac{(1-b) \left[ R_m(t) - \xi \right]^2}{(f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m))^2} \right| \\ &\leq \mu^2 \left| b \left[ \overline{F}_{T,n}(t) e^{t\xi} (R_m(t) - \xi) \right]^2 \right| + \nu^2 \left| (1-b) \left[ R_m(t) - \xi \right]^2 \\ &\leq (\mu^2 e^{t\xi} + \nu^2) (R_m(t) - \xi)^2 := S_2(t). \end{aligned}$$

Since the function  $S_2$  is continuous on the compact interval [0, 1], then it has a maximum  $\psi_2$ . Then

$$U_{2,k}(t,b;\theta) \leq \psi_2$$

Then  $U_{2,k}$  is dominated by an integrable function with respect to the density g for all  $(t, b) \in [0, 1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(i)

(iii) For k = 0, ..., m,

$$\begin{aligned} U_{3,k}(t,b;\theta) &= \left| \frac{\partial^2 \log(h(t,b;\theta))}{\partial \xi \partial \phi_k} \right| \\ &= \left| \frac{\partial h_{2,k}(t,b;\theta)}{\partial \xi} \right| \\ &\leq \mu^2 \left| \left[ e^{t\xi} (\xi R_m(t) - 1 - \xi^2) \right] \left[ f_{Z,n}(t) - \overline{F}_{T,n}(t) e^{t\xi} (f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m)) \right] \right| \\ &+ \mu^2 \left| \left[ -\overline{F}_{T,n}(t) f_{C,m}(t|\overline{\phi}_m) \xi e^{t\xi} + \overline{F}_{T,n}(t) \overline{F}_{C,m}(t|\overline{\phi}_m) (e^{t\xi} + \xi^2 e^{t\xi}) \right] \left[ e^{t\xi} R_m(t) - \xi e^{t\xi} \right] \right| \\ &+ \nu^2 \left| f_{C,m}(t|\overline{\phi}_m) - \xi \overline{F}_{C,m}(t|\overline{\phi}_m) \right| + \nu^2 \left| \overline{F}_{C,m}(t|\overline{\phi}_m) (R_m(t) - \xi) \right| \\ &\leq \mu^2 \left[ e^{t\xi} (\xi R_m(t) + 1 + \xi^2) \right] \times \left[ \frac{K_{max}}{h} + e^{t\xi} R_m(t) + \xi e^{t\xi} \right] \\ &+ \mu^2 \left[ e^{t\xi} (\xi R_m(t) + 1 + \xi^2) \right] \left[ e^{t\xi} R_m(t) + \xi e^{t\xi} \right] + \nu^2 \left[ R_m(t) + \xi \right]^2 := S_3(t). \end{aligned}$$

Since the function  $S_3$  is continuous on the compact interval [0, 1], then it has a maximum  $\psi_3$ . Then

$$U_{3,k}(t,b;\theta) \leq \psi_3.$$

Then  $U_{3,k}$  is dominated by an integrable function with respect to the density g for all  $(t, b) \in [0, 1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(iv) For  $k, j = 0, \dots m$  such that  $k \neq j$ ,

$$U_{4,j,k}(t,b;\theta) = \left| \frac{\partial^2 \log(h(t,b;\theta))}{\partial \phi_j \partial \phi_k} \right|$$
$$= \left| \frac{\partial h_{2,k}(t,b;\theta)}{\partial \phi_j} \right|.$$

Similarly to  $U_{2,k}$ ,  $U_{4,k}$  is dominated by an integrable function with respect to the density g for all  $(t,b) \in [0,1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(v) For  $k, \ldots m$ ,

$$U_{5,k}(t,b;\theta) = \left| \frac{\partial \log(h(t,b;\theta))}{\partial \xi} \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right|$$
$$= \left| \frac{\partial \log(h(t,b;\theta))}{\partial \xi} \right| \left| \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right|$$
$$= \left| h_1(t,b;\theta) \right| \left| h_{2,k}(t,b;\theta) \right|.$$

Or

$$\begin{aligned} |h_{1}(t,b;\theta)| &\leq \left| -\frac{b\overline{F}_{T,n}(t)\left[\xi e^{t\xi}f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(e^{t\xi} + \xi^{2}e^{t\xi})\right]}{f_{Z,n}(t) - \overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_{m}) - \xi\overline{F}_{C,m}(t|\overline{\phi}_{m})\right)} \right| \\ &+ \left| \frac{\left(1-b\right)\left[\xi f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2})\right]}{\left(f_{C,m}(t|\overline{\phi}_{m}) - \xi\overline{F}_{C,m}(t|\overline{\phi}_{m})\right)} \right|. \end{aligned}$$

It follows that

$$\begin{aligned} |h_{1}(t,b;\theta)| &\leq \mu \left| \xi e^{t\xi} f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(e^{t\xi} + \xi^{2}e^{t\xi}) \right| \\ &+ \nu \left| \xi f_{C,m}(t|\overline{\phi}_{m}) - \overline{F}_{C,m}(t|\overline{\phi}_{m})(1+\xi^{2}) \right| \\ &\leq \mu \left[ \xi e^{t\xi} f_{C,m}(t|\overline{\phi}_{m}) + (e^{t\xi} + \xi^{2}e^{t\xi}) \right] + \nu \left[ \xi f_{C,m}(t,\overline{\phi}_{m}) + (1+\xi^{2}) \right] \\ &= \mu \left[ \xi e^{t\xi} R_{m}(t) + (e^{t\xi} + \xi^{2}e^{t\xi}) \right] + \nu \left[ \xi R_{m}(t) + (1+\xi^{2}) \right] := S_{5,1}(t). \end{aligned}$$

Since the function  $S_{5,1}$  is continuous on the compact interval [0,1], then it has a maximum  $\psi_{5,1}$ . Then

$$|h_1(t,b;\theta)| \leq \psi_{5,1} \tag{8.8}$$

On the other hand, we have

$$\begin{aligned} |h_{2,k}(t,b;\theta)| &\leq \left| -\frac{b\overline{F}_{T,n}(t)e^{t\xi}\left(R_m(t)-\xi\right)}{f_{Z,n}(t)-\overline{F}_{T,n}(t)e^{t\xi}\left(f_{C,m}(t|\overline{\phi}_m)-\xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)} \right. \\ &+ \left| \frac{\left(1-b\right)\left(R_m(t)-\xi\right)}{\left(f_{C,m}(t|\overline{\phi}_m)-\xi\overline{F}_{C,m}(t|\overline{\phi}_m)\right)} \right| \\ &\leq \mu e^{t\xi}\left|R_m(t)-\xi\right|+\nu\left|R_m(t)-\xi\right| := S_{5,2}(t). \end{aligned}$$

Since the function  $S_{5,2}$  is continuous on the compact interval [0,1], then it has a maximum  $\psi_{5,2}$ . Then

$$|h_{2,k}(t,b;\theta)| \leq \psi_{5,2}.$$
 (8.9)

It follows from Equations (8.8) and (8.9) that

 $U_{5,k}(t,b;\theta) \leq \psi_{5,1}\psi_{5,2}.$ 

Then  $U_{5,k}$  is dominated by an integrable function with respect to the density g for all  $(t, b) \in [0, 1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(vi) For k = 0, ..., m,

$$U_{6,k}(t,b;\theta) = \left| \left[ \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right]^2 \right|$$
$$= \left| \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right| \left| \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right|$$
$$= \left| h_{2,k}(t,b;\theta) \right|^2.$$

It follows from Equation (8.9) that  $U_{6,k}$  is dominated by an integrable function with respect to the density g for all  $(t,b) \in [0,1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(vii) For  $k, j = 0, \dots m$  such that  $k \neq j$ ,

$$U_{7,k,j}(t,b;\theta) = \left| \frac{\partial \log(h(t,b;\theta))}{\partial \phi_j} \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right|$$
$$= \left| \frac{\partial \log(h(t,b;\theta))}{\partial \phi_j} \right| \left| \frac{\partial \log(h(t,b;\theta))}{\partial \phi_k} \right|$$
$$= \left| h_{2,j}(t,b;\theta) \right| \left| h_{2,k}(t,b;\theta) \right|$$
$$\leq \left[ (e^{t\xi} + 1)R_m(t) + 2\xi \right]^2.$$

Similarly to  $U_{6,k}$ ,  $U_{7,j,k}$  is dominated by an integrable function with respect to the density g for all  $(t,b) \in [0,1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

(viii)

$$U_8(t,b;\theta) = \left| \left[ \frac{\partial \log(h(t,b;\theta))}{\partial \xi} \right]^2 \right|$$
  
=  $\left| \frac{\partial \log(h(t,b;\theta))}{\partial \xi} \right| \left| \frac{\partial \log(h(t,b;\theta))}{\partial \xi} \right|$   
=  $|h_1(t,b;\theta)|^2$ .

It follows from Equation (8.8) that  $U_8$  is dominated by an integrable function with respect to the density g for all  $(t, b) \in [0, 1]^2$  and for all  $\theta \in \mathbb{R}^{m+2}$ .

Then condition (5) is satisfied.

Base on assumption  $(\mathbf{H}_4)$ , condition (6) of [60] is satisfied. Then the conditions (1)-(6) of Theorem 3.2 of [60] are verified, and the proof is complete.

#### **Proof of Proposition 7:**

Recall that

$$\theta = (\theta_1, \theta_2, \dots, \theta_{m+1}, \theta_{m+2}) = (\phi_0, \dots, \phi_m, \xi),$$
  
$$\theta_n = (\theta_{1,n}, \theta_{2,n}, \dots, \theta_{m+1,n}, \theta_{m+2,n}) = (\phi_{0,n}, \dots, \phi_{m,n}, \xi_n),$$

and

$$\widetilde{P}_{n,m}(t,s) = \begin{cases} \overline{F}_{C,n,m}(s)\overline{F}_{T,n}(t)^{1-\alpha_n(t)} & \text{if } \overline{F}_T(t)^{\alpha(t)} < \overline{F}_C(s)^{\beta(s)} \\ \overline{F}_{T,n}(t)\overline{F}_{C,n,m}(s)^{1-\beta_{n,m}(s)} & \text{if } \overline{F}_T(t)^{\alpha(t)} > \overline{F}_C(s)^{\beta(s)} \end{cases}$$

Based on Theorem 2.2 of [60],  $\theta_n \xrightarrow[n \to +\infty]{a.s} \theta$ , then

$$(\phi_{0,n},\ldots,\phi_{m,n}) \xrightarrow[n \to +\infty]{a.s} (\phi_0,\ldots,\phi_m)$$
 (8.10)

and

$$\xi_n \xrightarrow[n \to +\infty]{a.s} \xi. \tag{8.11}$$

Based on Equation (8.10), we have

$$\overline{F}_{C,n,m}(s) \xrightarrow[n \to +\infty]{a.s.} \overline{F}_{C,m}(s) \quad \text{and} \quad \beta_{n,m}(s) \xrightarrow[n \to +\infty]{a.s.} \beta_m(s).$$
(8.12)

On the other hand, we have

$$\left|\overline{F}_{C,n,m}(s) - \overline{F}(s)\right| \le \left|\overline{F}_{C,n,m}(s) - \overline{F}_{m}(s)\right| + \left|\overline{F}_{m}(s) - \overline{F}(s)\right|,$$

and

$$|\beta_{n,m}(s) - \beta(s)| \le |\beta_{n,m}(s) - \beta_m(s)| + |\beta_m(s) - \beta(s)|$$

Next, based on Theorem 1.1 in [8] and Theorem 1 in [29] in Section VII.2, we note that for *m* large enough,  $\lim_{m\to\infty} \overline{F}_{C,m}(s) = \overline{F}_C(s)$  and  $\lim_{m\to\infty} \beta_m(s) = \beta(s)$ . Then based on Equation (8.12),

$$\overline{F}_{C,n,m}(s) \xrightarrow[n,m\to+\infty]{a.s} \overline{F}_C(s) \quad \text{and} \quad \beta_{n,m}(s) \xrightarrow[n,m\to+\infty]{a.s} \beta(s).$$
(8.13)

According to Lemma 2 of [12], we have

$$\overline{F}_{T,n}(t) \xrightarrow[n \to +\infty]{a.s} \overline{F}_T(t) \quad \text{and} \quad \alpha_n(t) \xrightarrow[n \to +\infty]{a.s} \alpha(t).$$
 (8.14)

Then based on Equations (8.13) and (8.14), we obtain  $\lim_{n,m\to\infty} \widetilde{P}_{n,m}(t,s) = \widetilde{P}(t,s)$ .

On the other hand, based on Glivenko-Cantelli Theorem, we have

$$\overline{H}_n(t) \xrightarrow[n \to +\infty]{n \to +\infty} \overline{H}(t).$$
(8.15)

Then based on Equations (8.11) and (8.15), we have  $\lim_{n \to \infty} \tau_n = \tau$ .

**Conflict of Interest:** The authors declare that they have no conflicts of interest regarding the publication of this manuscript.

## References

- Aabring, J. and Van den Berg, G.J., (2003). The Identifiability of the Mixed Proportional Hazards Competing Risks Model. *Journal of the Royal Statistical Society Series B*, 65 701– 710.
- [2] Aalen, O., Borgan, O., and Gjessing, H., (2008). Survival and event history analysis: a process point of view. Springer Science and Business Media.
- [3] Abbring, J. H., (2012). Mixed Hitting-Time Models. Econometrica, 80 783–819.
- [4] Abbring, J. H., and Yu, Y., (2015). The empirical content of synchronization games. Working paper.
- [5] An, M. Y., Christensen, B. J., and Gupta, N. D., (2004). Multivariate mixed proportional hazard modelling of the joint retirement of married couples. *Journal of Applied Econometrics*, 19 687–704.
- [6] Andersen, P. K., Borgan, O., Gill, R. D., and Keiding, N., (2012). Statistical models based on counting processes. Springer Science and Business Media.
- [7] Arnold, B. C., and Brockett, P. L., (1983). Identifiability for dependent multiple decrement/competing risk models. Scandinavian Actuarial Journal, 2 117–127.
- [8] Babu, G. J., Canty, A. J., and Chaubey, Y.P., (2002). Application of Bernstein Polynomials for smooth estimation of a distribution and density function. *Journal of Statistical Planning and Inference.* 105 377—392.
- [9] Bielecki, T. R., and Rutkowski, M., (2013). Credit risk: modeling, valuation and hedging. Springer Science and Business Media.

- [10] Braekers, R. and Veraverbeke, N., (2005). A copula-graphic estimator for the conditional survival function under dependent censoring. *Canadian Journal of Statistics*, **33**, 429–447.
- [11] Breslow, N. and Crowley, J., (1974). A large sample study of the life table and product limit estimates under random censorship. *The Annals of statistics*, **2**, 437–453.
- [12] Cai, Z., (1997). Asymptotic properties of Kaplan-Meier estimator for censored dependent data. Statistics and probability letters, 37, 381–389.
- [13] Cherubini, U., Durante, F., and Mulinacci, S., (2015). Marshall–Olkin distributions–advances in theory and applications. Springer Proceedings in Mathematics and Statistics, Springer International Publishing.
- [14] Choi, J. P., (1991). Dynamic R and D competition under hazard rate uncertainty. The RAND Journal of Economics, 596–610.
- [15] Chen, Y.-H., (2010). Semiparametric marginal regression analysis for dependent competing risks under an assumed copula. *Journal of the Royal Statistical Society: Series B*, **72**, 235–251.
- [16] Coile, C., (2004). Health shocks and couples' labor supply decisions.
- [17] Czado, C. and Van Keilegom, I., (2022). Dependent censoring based on parametric copulas. Biometrika, asac067.
- [18] de Unã-Àlvarez, J. and Veraverbeke, N., (2017). Copula-graphic estimation with lefttruncated and right-censored data. *Statistics*, **51**, 387–403.
- [19] De Paula, A., (2009). Inference in a synchronization game with social interactions. Journal of econometrics. Journal of econometrics, 148, 56–71.
- [20] Deresa, N. W., and Van Keilegom, I., (2021). On semiparametric modelling, estimation and inference for survival data subject to dependent censoring. *Biometrika*, **108**, 965–979.
- [21] Deresa, N. W., Van Keilegom, I., and Antonio, K., (2022). Copula-based inference for bivariate survival data with left truncation and dependent censoring. *Insurance: Mathematics and Economics*, 1–21.
- [22] Deresa, N. W. and Van Keilegom, I., (2023). Copula based Cox proportional hazards models for dependent censoring. *Journal of the American Statistical Association*, 1–11.
- [23] Duran, A. J., (1991). A bound on the Laguerre polynomials. Studia Mathematica, 100, 169– 181.
- [24] Ebrahimi, N., Molefe, D., and Ying, Z., (2003). Identifiability and censored data. *Biometrika*, 90, 724–727.
- [25] Embrechts P., Lindskog F., McNeil A.J., (2003) Modelling dependence with copulas and applications to risk management. S. Rachev (Ed.), Handbook of Heavy Tailed Distributions in Finance, Elsevier, 329—384
- [26] Emura, T., and Chen, Y. H., (2018). Analysis of Survival Data with Dependent Censoring: Copula-Based Approaches. Singapore: Springer.

- [27] Escobar-Bach, M., and Helali, S., (2024). Dependent censoring with simultaneous death times based on the Generalized Marshall-Olkin model. *Journal of Multivariate Analysis*, **105347**.
- [28] Farzin, Y. H., Huisman, K. J., and Kort, P. M., (1998). Optimal timing of technology adoption. Journal of Economic Dynamics and Control, 22, 779–799.
- [29] Feller, W., (1991). An introduction to probability theory and its applications. John Wiley and Sons.
- [30] Heckman, J. J. and Honore, B.E., (2013). The identifiability of the competing risks model. Biometrika, 79, 325—330.
- [31] Heckman, J. J., and Navarro, S., (2007). Dynamic discrete choice and dynamic treatment effects. *Journal of Econometrics*, **136**, 341–396.
- [32] Hofert, M., and Vrins, F., (2013). Sibuya copulas. Journal of Multivariate Analysis, , 318–37.
- [33] Honor, B. E., and De Paula, Á., (2010). Interdependent durations. The Review of Economic Studies, 77, 1138–1163.
- [34] Honor, B. E., and De Paula, A., (2018). A new model for interdependent durations. Quantitative Economics, 9, 1299–1333.
- [35] Kakizawa, Y., (2004). Bernstein polynomial probability density estimation. Journal of Nonparametric Statistics. 16, 709–729.
- [36] Kleinbaum, D. G., and Klein, M., (1996). Survival analysis a self-learning text. Springer.
- [37] Kreiss, A. and Van Keilegom, I., (2022). Semi-parametric estimation of incubation and generation times by means of Laguerre polynomials. *Journal of Nonparametric Statistics.* 34, 570–606.
- [38] Lancaster, T., (1972). A stochastic model for the duration of a strike. Journal of the Royal Statistical Society: Series A (General), 135, 257–271.
- [39] Lancaster, T., (1979). Econometric methods for the duration of unemployment. Econometrica: Journal of the Econometric Society, 939–956.
- [40] Leblanc, A., (2012). On estimating distribution functions using Bernstein polynomials. Annals of the Institute of Statistical Mathematics, 64, 919–943.
- [41] Lee, S., (2006). Identification of a competing risks model with unknown transformations of latent failure times. *Biometrika*, 93, 996–1002.
- [42] Lee, S. and Lewbel, A., (2013). Nonparametric identification of accelerated failure time competing risks models. *Econometric Theory*, 29, 905—919
- [43] Li, X., and Pellerey, F., (2011). Generalized Marshall–Olkin distributions and related bivariate aging properties. *Journal of Multivariate Analysis*, 102, 1399–1409.
- [44] Liu, R., (2020). A competing risks model with time-varying heterogeneity and simultaneous failure. Quantitative Economics, 11, 535–577.
- [45] Maller, R. A. and Zhou, S., (1992). Estimating the proportion of immunes in a censored sample. Biometrika, 79, 731–739.

- [46] Marshall A.W. and Olkin I., (1967). A multivariate exponential distribution. Journal of the American Statistical Association, 62, 30–44.
- [47] Richard, J., (1982). Adoption and Diffusion of an Innovation of Uncertain Profitability. Journal of Economic Theory, 27, 182–93.
- [48] Reinganum, J. F., (1982). A dynamic game of R and D: Patent protection and competitive behavior. *Econometrica: Journal of the Econometric Society*, 671–688.
- [49] Ryu, K., (1993). Structural duration analysis of management data. Journal of Econometrics, 57,91–115.
- [50] Scharfstein, D. O., Rotnitzky, A., and Robins, J. M., (2003). Adjusting for nonignorable dropout using semiparametric nonresponse models. *Journal of the American Statistical Association*, 94, 1096–1120.
- [51] Schönbucher, P. J., (2003). Credit derivatives pricing models: models, pricing and implementation. John Wiley and Sons.
- [52] Shorack, G. R. and Wellner, J. A., (1986). Empirical Processes with Applications to Statistics. John Wiley and Sons, Inc. New York.
- [53] Sibuya, M., (1960). Bivariate extreme statistics. Annals of the Institute of Statistical Mathematics. 11, 195–210.
- [54] Sklar, M., (1959). Fonctions de répartition à n dimensions et leurs marges. Publications de l'Institut de Statistique de l'Universite de Paris, 8, 229–231.
- [55] Sun, Y., Mendoza-Arriaga, R., and Linetsky, V., (2017). Marshall–Olkin distributions, subordinators, efficient simulation, and applications to credit risk. Advances in Applied Probability, 49, 481–514.
- [56] Tsiatis A., (1975). A nonidentifiability aspect of the problem of competing risks. *Proceedings* of the National Academy of Sciences, **72**, 20–22.
- [57] Van Keilegom, I., and Kekeç, E., (2024). Estimation of the density for censored and contaminated data. Stat, 13, 2e651.
- [58] Vitale, R. A., (1975). A Bernstein polynomial approach to density function estimation. In Statistical inference and related topics . 87–99.
- [59] Vrins, F. D., (2010). Analytical pricing of basket default swaps in a dynamic Hull and White framework SSRN 1590932.
- [60] White, H., (1982). Maximum likelihood estimation of misspecified models. Stat, 1, 1–25.
- [61] Zheng M. and J. P. Klein, (1995). Estimates of marginal survival for dependent competing risks based on an assumed copula. *Biometrika*, 82, 127–138.